Author’s response to reviews

Title: Mental health nurses' attitudes, experience, and knowledge regarding routine physical healthcare: Systematic, integrative review of studies involving 7,549 nurses working in mental health settings

Authors:
Dickens - Geoffrey L. (g.dickens@westernsydney.edu.au)
Robin Ion (r.ion@abertay.ac.uk)
Cheryl Waters (cheryl.d.waters@gmail.com)
Evan Atlantis (e.atlantis@westernsydney.edu.au)
Bronwyn Everett (b.everett@westernsydney.edu.au)

Version: 2 Date: 11 Mar 2019

Author’s response to reviews:

Reviewer reports:

Pamela Kirkpatrick (Reviewer 3): All aspects from the initial peer review have been addressed adequately.

This is a robust review focused on a specific aspect of mental health nursing.

It competently addresses the remit set out in the objectives and sets the future research direction.

Some minor areas requiring final editing:

P.2 line 17 - there is an error in the abstract where the inclusion criteria are noted as i), ii) and iii)? Something is missing and it therefore lacks consistency between abstract and introduction.

- Amended

p.8 line 57 - reference source should be listed.

- Amended
P.19 line 47 - error or unfinished sentence.
- Amended

P.21 line 50 (studies spelling error)
- Amended

P.31 line 33 & P.32 L18, spacing errors. Others throughout the reference list.
- Amended and checked thoroughly